Author's response to reviews

Title: Behaviour change in perinatal care practices among rural women exposed to a women's group intervention in Nepal [ISRCTN31137309]

Authors:

   Angela M Wade (awade@ich.ucl.ac.uk)
   David Osrin (d.osrin@ich.ucl.ac.uk)
   Bhim P Shrestha (mirajnk@ntc.net.np)
   Aman Sen (miramak@ntc.net.np)
   Joanna Morrison (j.morrison@ich.ucl.ac.uk)
   Kirti M Tumbahangpe (miramak@ntc.net.np)
   Dharma S Manandhar (ds@healthnet.org.np)
   Anthony ML Costello (a.costello@ich.ucl.ac.uk)

Version: 4  Date: 12 June 2006

Author's response to reviews: see over
Title: Behaviour change in perinatal care practices among rural women exposed to a women’s group intervention in Nepal

Response to formatting changes requested June 2006:

1. Country - Please remember to include the country in the affiliation details.
   - These have been added where previously omitted.

2. Qualifications - Please remove the authors’ qualifications (e.g. PhD, M.D.) from the title page.
   - We have removed these.

3. Please delete page 3 of the manuscript (Key Messages).
   - Done

4. Introduction - Please rename this section ‘Background’.
   - Done

5. References - References should be cited in the text using consecutive normal case numbers in square brackets e.g. [1]. Please do not use superscript numbers.
   - Edited as requested.

6. Please include a conclusions sub heading.
   - Added

7. References : The reference list should contain all authors’ names, regardless of how many there are. The term ‘et al’ should not be used.
   - Edited as requested.

8. Please move competing interests, authors' contributions and acknowledgements section to before the reference list.
   - Done

9. Please delete the additional file.
   - Done